\documentclass[english,11pt]{article}
\usepackage[latin1]{inputenc}
\usepackage{amsmath,amsthm,amssymb,babel}
%\usepackage[notref,notcite]{showkeys}
\renewcommand{\arraystretch}{1.7}
\textwidth145mm 
\textheight195mm 
%\hoffset-24mm \voffset-20mm
\newtheorem{teo}{Theorem}
\newtheorem{defin}{Definition}
\newtheorem{proposition}{Proposition}
\newtheorem{prop}{Proposition}
%\newtheorem{Defi}[theo]{Definition}
\newtheorem{cor}{Corollary}
\newtheorem{lemma}{Lemma}
\newtheorem{rem}{Remark}
%\renewcommand{\arraystretch}{1.7}
%\textwidth172mm \textheight22cm \hoffset-24mm \voffset-20mm
\font\sans=cmss10
\def\proof{{\it Proof.}\ }
\def\endproof{\hfill $\Box$\par\vskip3mm}
\def\pt#1{{\it Proof of Theorem \ref{#1}.}}
\def\eq#1{(\ref{#1})}
\def\th#1{Theorem \ref{#1}}
\def\neweq#1{\begin{equation}\label{#1}}
\def\endeq{\end{equation}}
\def\weak{\rightharpoonup}
\def\ep{\varepsilon}
\def\half{{1\over2}}
\def\phi{\varphi}
\def\RR{{\mathbb R} }
\def\NN{{\mathbb N} }
\def\ZZ{{\mathbb Z} }
\def\supp{{\rm supp}}
\def\di{\displaystyle}
\def\ri{\rightarrow}
\def\intom{\int_\Omega}
\date{}

\title{ Small perturbations of nonlocal biharmonic problems with variable exponent and competing nonlinearities}
\author{ Khaled Kefi$\,^{1,2}$ and Vicen\c tiu D. R\u adulescu$\,^{3,4}$\footnote{Corresponding author} \\
$^1\,${\footnotesize Community College of Rafha, Northern Border University, Kingdom of Saudi Arabia}\\
$^2\,${\footnotesize Department of Mathematics, Faculty of Sciences Tunis El Manar, 1060 Tunis, Tunisia}\\
 {\footnotesize E-mail: {\tt khaled\_kefi@yahoo.fr} }\\
 $^3\,${\footnotesize Faculty of Applied Mathematics, AGH University of Science and Technology,} \\ {\footnotesize al. Mickiewicza 30, 30-059 Krak\'ow, Poland}\\
$^4\,${\footnotesize Department of Mathematics, University of Craiova, 
          200585 Craiova, Romania}\\
 {\footnotesize E-mail: {\tt vicentiu.radulescu@imar.ro} }}
\begin{document}
\begin{thebibliography}{99}

\bibitem{AN} S.N. Antontsev and S.I. Shmarev, A model porous medium equation with variable exponent of nonlinearity: existence,
uniqueness and localization properties of solutions, {\it Nonlinear Anal.} {\bf 60} (2005), 515-545.

\bibitem{AR} A. Arosio and S. Panizzi, On the well-posedness of the Kirchhoff string, {\it Trans. Amer. Math. Soc.} {\bf 348} (1996), 305-330.

\bibitem{SV} S. Baraket and V. R\u adulescu, Combined effects of concave-convex nonlinearities in a fourth-order problem with variable exponent,
{\it Adv. Nonlinear Stud.} {\bf 16} (2016), 409-419.

\bibitem{mingi1} P. Baroni, M. Colombo, and G. Mingione, Nonautonomous functionals, borderline cases and related function classes, {\it Algebra i Analiz} {\bf 27} (2015), no. 3, 6-50; translation in {\it St. Petersburg Math. J.} {\bf 27} (2016), no. 3, 347-379.

\bibitem{KM} K. Benali and M. Bezzarga, On a nonhomogeneous quasilinear eigenvalue problem in Sobolev
spaces with variable exponent, in {\it Potential Theory and Stochastics
in Albac} Aurel Cornea Memorial Volume, Conference Proceedings,
Albac, September 4-8, 2007, Theta 2008, 21-34.

\bibitem{MK} M. Bouslimi and K. Kefi, Existence of solution for an indefinite weight quasilinear problem with variable exponent, {\it Complex Variables and Elliptic Equations} {\bf 58} (2013), 1655-1666.
    
\bibitem{R00} Y. Chen, S. Levine, and M. Rao, Variable exponent, linear growth functionals in image processing, {\it SIAM J. Appl. Math.} {\bf 66} (2006), 1383-1406.

\bibitem{patrizia1} F. Colasuonno and P. Pucci, Multiplicity of solutions for $p(x)$-polyharmonic elliptic Kirchhoff equations, {\it Nonlinear Anal.} {\bf 74} (2011), no. 17, 5962-5974.

 \bibitem{mingi02} M. Colombo and G. Mingione, Regularity for double phase variational problems, {\it Arch. Ration. Mech. Anal.} {\bf 215} (2015), no. 2, 443-496.

    \bibitem{mingi2} M. Colombo and G. Mingione, Bounded minimisers of double phase variational integrals, {\it Arch. Ration. Mech. Anal.} {\bf 218} (2015), 219-273.

\bibitem{CO1} F.J.S.A. Corr\^{e}a and G.M. Figueiredo, On an $p$-Kirchhoff equation via Krasnoselskii's genus, {\it Appl. Math. Letters} {\bf 22} (2009), 819-822.

\bibitem{DR}  M. Dreher, The wave equation for the $p$-Laplacian, {\it Hokkaido Math. J.} {\bf 36} (2007), 21-52.

\bibitem{R7} D. Edmunds and J. Rakosnik, Sobolev embeddings with variable exponent, {\it Studia Math.} {\bf{143}} (2000), 267-293.

\bibitem{EK} I. Ekeland, On the variational principle, {\it J. Math. Anal. Appl.} \textbf{47} (1974), 324-353.
    
\bibitem{eleuteri} M. Eleuteri, P. Marcellini, and E. Mascolo, Lipschitz continuity for energy integrals with variable exponents, {\it Atti Accad. Naz. Lincei Rend. Lincei Mat. Appl.} {\bf 27} (2016), no. 1, 61-87.

\bibitem{FH} X. Fan and X. Han, Existence and multiplicity of solutions for $p(x)$-Laplacian equations in ${\mathbb R}^N$, {\it Nonlinear Anal} {\bf 59} (2004), 173-188.

\bibitem{FW} A. Ferrero and G. Warnault, On solutions of second and fourth order elliptic equations with power-type nonlinearities, {\it Nonlinear Anal.} {\bf 70} (2009), 2889-2902.
    
\bibitem{fuy} Y. Fu, and Y. Shan, On the removability of isolated singular points for elliptic equations involving variable exponent, {\it Adv. Nonlinear Anal.} {\bf 5} (2016), no. 2, 121-132.

    \bibitem{halsey} T.C. Halsey, Electrorheological fluids, {\it Science} {\bf 258} (1992), 761-766.

    \bibitem{harj} P. Harjulehto, P. H\"ast\"o, U.V. Le, and M. Nuortio, Overview of differential equations with non-standard growth, {\it Nonlinear Anal.} {\bf  72} (2010), 4551-4574.
        
\bibitem{heidari} M.R. Heidari Tavani, G.A. Afrouzi, S. Heidarkhani, Multiplicity results for perturbed fourth-order Kirchhoff-type problems, {\it Opuscula Math.} {\bf 37} (2017), no. 5, 755-772.

\bibitem{KK} K. Kefi, $p(x)$-Laplacian with indefinite weight, {\it Proc. Amer. Math. Soc.} {\bf 139} (2011), 4351-4360.
    
\bibitem{KV}  K. Kefi and V. R\~adulescu,  On a p(x)-biharmonic problem with singular weights, {\it Z. Angew. Math. Phys.} {\bf 68} (2017), no. 4, Art. 80, 13 pp.

\bibitem{K}  G. Kirchhoff, {\it Mechanik}, Teubner, Leipzig, Germany, 1883.

\bibitem{LION} J.-L. Lions, On some questions in boundary value problems of mathematical physics, in
 {\it Proceedings of International Symposium on Continuum Mechanics and Partial Differential Equations}, Rio de Janeiro 1977  (de la Penha, Medeiros, Eds.), {\it Math. Stud., North Holland} {\bf 30} (1978), 284-346.

\bibitem{marce1} P. Marcellini, On the definition and the lower semicontinuity of certain quasiconvex integrals, {\it Ann. Inst. H. Poincar\'e, Anal. Non Lin\'eaire} {\bf 3} (1986), 391-409.

\bibitem{marce2} P. Marcellini, Regularity and existence of solutions of elliptic equations with $(p, q)$-growth conditions, {\it J. Differential Equations} {\bf 90} (1991), 1-30.

\bibitem{R0} R.A. Mashiyev, S. Ogras, Z. Yucedag, and M. Avci, Existence and multiplicity of weak solutions for nonuniformly elliptic equations with non-standard growth condition, {\it Complex Variables and Elliptic Equations} {\bf 57} (2012),  579-595.

\bibitem{M1} T.G. Meyers, Thin film with hight surface tension, {\it Siam Rev.} {\bf 40} (1998), 441-462.

\bibitem{molica} G. Molica Bisci and D. Repov\v{s},  On doubly nonlocal fractional elliptic equations, {\it Atti Accad. Naz. Lincei Rend. Lincei Mat. Appl.} {\bf 26} (2015), no. 2, 161-176.

\bibitem{orlicz} W. Orlicz, \"Uber konjugierte Exponentenfolgen, {\it Studia Math.} {\bf
3} (1931), 200-212.

\bibitem{patrizia02} P. Pucci, M. Xiang, and B. Zhang, Existence and multiplicity of entire solutions for fractional $p$-Kirchhoff equations, {\it Adv. Nonlinear Anal.} {\bf 5} (2016), no. 1, 27-55.

\bibitem{puccicv} P. Pucci, M. Xiang, and B. Zhang, Multiple solutions for nonhomogeneous Schr\"odinger-Kirchhoff type equations involving the fractional $p$-Laplacian in $\RR^N$, {\it Calc. Var. Partial Differential Equations} {\bf 54} (2015),  2785-2806.
    
\bibitem{patrizia2} P. Pucci and Q. Zhang, Existence of entire solutions for a class of variable exponent elliptic equations, {\it J. Differential Equations} {\bf 157} (2014),  1529-1566.

\bibitem{radnla} V.D. R\u adulescu, Nonlinear elliptic equations with variable exponent: old and new, {\it Nonlinear Anal.} {\bf 121} (2015), 336-369.

\bibitem{radrep} V.D. R\u adulescu and D.D. Repov\v{s}, {\it Partial Differential Equations with Variable Exponents: Variational Methods and Qualitative Analysis}, Monographs and Research Notes in Mathematics, Taylor \& Francis, Chapman and Hall/CRC,  2015.

\bibitem{ZF} A. Zang and Y. Fu, Interpolation inequalities for derivatives in variable exponent Lebesgue Sobolev spaces, {\it Nonlinear Anal.} {\bf 69} (2008), 3629-3636.

\bibitem{zh2} V.V. Zhikov, Lavrentiev phenomenon and homogenization for some variational problems, {\it C. R. Acad. Sci. Paris S\'er. I Math.} {\bf 316} (1993), no. 5, 435-439.

\end{thebibliography}

\baselineskip16pt \maketitle
\noindent{\small{\bf{\sc Abstract.}}
The present paper deals with the analysis of combined effects of an absorption term and a small perturbation of the reaction term in a $p(x)$-biharmonic Kirchhoff problem with Navier boundary condition.
The main result in this work establishes the existence of a continuous spectrum consisting in an interval.  The proofs combine variational methods with energy estimates.\\
{\bf 2010 Mathematics Subject Classification :} 35D05, 35D30, 35J58, 35J60, 35D30, 58E05.\\
\small{\bf Key words:}  $p(x)$-biharmonic Kirchhoff problem, Navier boundary condition, Ekeland
variational principle, generalized Sobolev spaces, continuous spectrum.}

\section{Introduction}
The qualitative analysis of nonlinear problems with one or several variable exponents started with the pioneering papers of Halsey \cite{halsey} and Zhikov \cite{zh2}, in strong relationship with the  behavior of strongly anisotropic materials. Their work is an important contribution to the refined mathematical analysis of nonlinear problems with one or more variable exponents, mainly because  it allows the understanding of some classes of nonlinear problems with possible lack of uniform convexity.
Nonlinear problems with this structure are motivated by numerous models in the applied sciences that are driven by partial differential equations with one or more variable exponents. In some circumstances, the standard analysis based on the theory of usual Lebesgue and Sobolev function  spaces,  $L^p$ and $W^{1,p}$, is not appropriate in the framework of material that involve non-homogeneities. For instance, both electro-rheological ``smart" fluids  and phenomena arising in image processing are described in a correct way by nonlinear models in which the exponent $p$ is not necessarily constant.
The variable exponent describe the geometry of a material which is allowed to change its hardening  exponent according to the point.
This leads to the analysis of variable exponents Lebesgue and Sobolev function spaces (denoted by $L^{p(x)}$ and $W^{1,p(x)}$), where $p$ is a real-valued (non-constant) function.  This is a common abstract framework in homogenization and nonlinear elasticity. We refer here to  the monograph by R\u adulescu and Repov\v s \cite{radrep}, which includes a thorough variational and topological analysis of several classes of problems with variable exponent (see also the survey paper R\u adulescu \cite{radnla} and the important contributions by Pucci {\it et al.} \cite{patrizia1, patrizia02, patrizia2}).

In this paper, we first discuss
the existence of a continuous spectrum consisting in an interval for the following nonlocal biharmonic problem with variable exponent:
\begin{equation}\label{p1}
\left\{\begin{array}{lll}M(t)(\Delta^{2}_{p(x)} u+a(x)|u|^{p(x)-2}u)&=\lambda
V_1(x)|u|^{q(x)-2}u,
\quad & \mbox{in}\ \Omega\\
%&u\not\equiv 0,\quad  &\mbox\in\Omega\\
&u=\Delta u=0,\quad   & \mbox{on}\ \partial\Omega,
\end{array}\right.
\end{equation}
where $\Delta^2_{p(x)} u = \Delta(|\Delta u|^{p(x)-2} \Delta u)$ is the
$p(x)$-biharmonic operator and $\Omega$ is a smooth bounded domain in $\mathbb{R}^{N}$. We assume that $M(t)$ is a continuous function and $$t:=\int_{\Omega}\frac{1}{p(x)}(|\Delta u|^{p(x)}dx+a(x)|u|^{p(x)})dx,$$
 $\lambda$ is a positive parameter, $p, q$ are continuous functions
on $\overline{\Omega}$ and $a\in L^{\infty}(\Omega)$ such that $\displaystyle {\rm essinf}_{x\in\Omega}a(x)>0$.
We assume that $V_1$ is an potential in a generalized Lebesgue space
such that $V_1>0$ in an open set $\Omega_{0}\subset\subset\Omega$, where $|\Omega_{0}| >0$.

Next, we focus on the following perturbed problem
\begin{equation}\label{1}
\left\{\begin{array}{lll}M(t)(\Delta^{2}_{p(x)} u+a(x)|u|^{p(x)-2}u)&=\lambda (V_1(x)|u|^{q(x)-2}u- V_2(x) |u|^{\alpha(x)-2}u)
\quad & \mbox{in}\ \Omega\\
%&u\not\equiv 0,\quad  &\mbox\in\Omega\\
&u=\Delta u=0,\quad   & \mbox{on}\ \partial\Omega,
\end{array}\right.
\end{equation}
where $ \alpha$ is a continuous function on $\overline{\Omega}$ and
 $V_2$ is a nonnegative one in a generalized Lebesgue spaces.

Problems (\ref{p1}) and \eqref{1} are {\it nonlocal problems} because of the presence of the term $M$, which implies that the equations in (\ref{p1})and \eqref{1} are no longer pointwise. This provokes some mathematical difficulties which make the study of such a problem particulary interesting.

In 1883, Kirchhoff \cite{K} introduced a model given by the equation
 \begin{equation}\label{eq1.2}
 \rho \frac{\partial^2 u}{\partial t^2} - \Big(\frac{\rho_0}{h}
 + \frac{E}{2L} \int_0^L |\frac{\partial u}{\partial x}|^{2}dx\Big)
 \frac{\partial^2 u}{\partial x^2} = 0,
\end{equation}
which extends the classical d'Alembert's wave equation by
considering the effects of the changes in the length of the
strings during the vibrations. The parameters in the above equation have the following meanings: $L$ is the length of the string, $h$ is the area of the cross-section, $E$ is the young modulus of the material, $\rho$ is the mass density and $\rho_0$ is the initial tension. A feature of problem (\ref{eq1.2}) is that the equation contains a nonlocal coefficient $\frac{\rho_0}{h}
 + \frac{E}{2L} \int_0^L |\frac{\partial u}{\partial x}|^{2}dx$, which depends on the average $\frac{1}{2L} \int_0^L |\frac{\partial u}{\partial x}|^{2}dx.$
 Nonlocal effects also find various applications in biological systems.

 After the work of Lions \cite{LION}, various equations of Kirchhoff type have been investigated, see \cite{AR, DR}. Moreover, Kirchhoff-type equations involving $p$-Laplacian and $p(x)$-Laplacian have been studied in many papers; see, \cite{CO1, eleuteri, fuy, heidari, molica}. A parabolic version
of problem \eqref{eq1.2} can be used to describe the growth and movement of a particular species.
The movement, modeled by the integral term, is assumed to be dependent on the energy of the
entire system with $u$
being its population density. Alternatively, the movement of  particular species may be subject to the total population density within the domain (for instance, the
spreading of bacteria), which gives rise to nonlocal parabolic equations. We refer to \cite{puccicv} for details.

 We also mention that fourth-order elliptic equations arise in many domains like micro-electro-mechanical systems, surface diffusion on solids, thin film theory, flow in Hele-Shaw cells and phase field models of multiphasic systems, see \cite{SV,FW,M1}. Recent contributions concerning a fourth order elliptic problems with $p(x)$ biharmonic operators can be found in \cite{KV}.

In the present paper, we study problem (\ref{p1}) under the following assumptions:
\\
$(H_0)$ $M:\mathbb{R}\rightarrow [m_0,+\infty)$ is a continuous function, with $m_0>0$;\\
$(H_1)$ there exists $0<\theta<1$ such that $$\widehat{M}(t)\geq (1-\theta)M(t)t\ \mbox{for all}\ t\geq 0,\ \mbox{where}\ \ \widehat{M}(t)=\int_{0}^{t} M(s)ds;$$
$(H_2)$ $1<q(x)<p(x)< \frac{N}{2}<s_1(x)$, $\mbox{for all }x\in \overline{\Omega}$, $V_1\in L^{s_1(x)}(\Omega)$ and $V_1>0 \ \mbox{in} \ \Omega_{0}\subset\subset \Omega$, with $\mid\Omega_{0}\mid>0$.

We point out that Kefi \cite{KK} was the first to introduce assumption like $(H_2)$ to study problems involving Lebesgue and Sobolev spaces with variable exponents.

\section{Terminology and abstract setting}
To study $p(x)$-biharmonic problems, we need some results on the spaces $L^{p(x)}(\Omega)$, $W^{1,p(x)}(\Omega)$ and $W^{k,p(x)}(\Omega)$; see \cite{harj,radnla,radrep} for details, complements and proofs.

 Let $$C_{+}(\overline{\Omega}):=\{h: h\in C(\overline{\Omega}), h(x)>1, \mbox{for all}\;\; x\in \overline{\Omega}\}.$$
For any $p\in C_{+}(\overline{\Omega})$, we denote $1< p^{-}:=\displaystyle\min_{x\in \overline{\Omega}}p(x) \leq  p^{+}=\displaystyle\max_{x\in \overline{\Omega}}p(x)<\infty$ and
 $$L^{p(x)}(\Omega)=\{u:\Omega\rightarrow \mathbb{R} \mbox{ measurable and } \int_{\Omega}|u(x)|^{p(x)}dx<\infty\}.$$
 The spaces $L^{p(x)}(\Omega)$ have been introduced by Orlicz \cite{orlicz}.

The space $L^{p(x)}(\Omega)$ is endowed with the  Luxemburg norm, which is defined by $$|u|_{p(x)}=\inf\left\{\mu>0:\int_{\Omega}\left|\frac{u(x)}{\mu}\right|^{p(x)}dx\leq 1\right\}.$$
Clearly, when $p(x)\equiv p$, the space $L^{p(x)}(\Omega)$ reduces to the classical Lebesgue space $L^p(\Omega)$ and the norm $|u|_{p(x)}$ reduces to the standard norm $\|u\|_{L^p}=(\int_{\Omega}|u|^p dx )^{\frac{1}{p}}$ in $L^p(\Omega)$.

For any positive integer $k$,  let $$W^{k,p(x)}(\Omega)=\{u\in L^{p(x)}(\Omega): D^{\alpha}u\in L^{p(x)}(\Omega), |\alpha|\leq k\},$$ where $\alpha=(\alpha_1,\alpha_2,\ldots ,\alpha_n) $ is a multi-index, $|\alpha|=\sum_{i=1}^{n}\alpha_i$ and
$D^{\alpha}u=\frac{\partial^{|\alpha|}u}{\partial^{\alpha_1}x_1\ldots\partial^{\alpha_N}x_N}.$
Then $W^{k,p(x)}(\Omega)$ is a separable and reflexive Banach space equipped with the norm $$\|u\|_{k,p(x)}=\displaystyle\sum_{|\alpha|\leq k}|D^{\alpha}u|_{p(x)}.$$

The space $W_{0}^{k,p(x)}(\Omega)$ is the closure of $C_{0}^{\infty}(\Omega)$ in $W^{k,p(x)}(\Omega).$

Let $L^{p'(x)}(\Omega)$ be the conjugate space of $L^{p(x)}(\Omega)$ with $\frac{1}{p}+\frac{1}{p'}=1$. Then the following H\"{o}lder-type inequality
\begin{equation}\label{7}
\left|\int_{\Omega}u v dx\right|\leq \left(\frac{1}{p^{-}}+\frac{1}{(p')^{-}}\right)|u|_{p(x)}|v|_{p'(x)}, \ \ \mbox{for all}\ u\in L^{p(x)}(\Omega), v\in L^{p'(x)}(\Omega)
\end{equation}
holds.
Moreover, if $h_1$, $h_2$ and
$h_3:\overline{\Omega}\rightarrow(1,\infty)$ are Lipschitz
continuous functions such that $1/h_{1}(x) + 1/h_{2}(x) + 1/h_{3}(x)
= 1$, then for any $u \in L^{h_{1}(x)}(\Omega), v\in
L^{h_{2}(x)}(\Omega)$ and $w\in L^{h_{3}(x)}(\Omega)$ the following
inequality holds (see \cite[Proposition 2.5]{FH}):
\begin{equation}\label{Hol1}
\left|\int_\Omega uvw\,dx\right|\leq\left(\frac{1}{h_{1}^-}+
\frac{1}{{h_{2}}^-}+\frac{1}{{h_{3}}^-}\right)|u|_{h_{1}(x)}|v|_{h_{2}(x)}|w|_{h_3(x)}.
\end{equation}
Inequality (\ref{7}) and its generalized version (\ref{Hol1}) are due to Orlicz \cite{orlicz}.

The modular on the space $L^{p(x)}(\Omega)$ is the map $\rho_{p(x)}:L^{p(x)}(\Omega)\rightarrow \mathbb{R}$ defined by $$\rho_{p(x)}(u):=\int_{\Omega}|u|^{p(x)}dx.$$

\begin{prop}\label{ll}(See \cite{R0})
For all $u, v\in L^{p(x)}(\Omega)$, we have
\begin{enumerate}
                       \item $|u|_{p(x)}<1\ (resp. =1, >1) \Leftrightarrow \rho_{p(x)}(u)<1\ (resp. =1, >1).$
                       \item $\min(|u|^{p^{-}}_{p(x)},|u|^{p^{+}}_{p(x)})\leq\rho_{p(x)}(u)\leq \max(|u|^{p^{-}}_{p(x)},|u|^{p^{+}}_{p(x)}).$
                       \item $\rho_{p(x)}(u-v)\rightarrow 0\Leftrightarrow |u-v|_{p(x)}\rightarrow 0.$
                     \end{enumerate}
\end{prop}
\begin{prop}\label{norm1}(See \cite{R7}) Let $p$ and $q$ be  measurable functions such
that $p\in L^{\infty}(\Omega)$, and $1\leq p(x)q(x)\leq\infty$, for a.e. $x\in\Omega$. Let $u\in L^{q(x)}(\Omega)$, $u\neq0$. Then
$$ \min(|u|^{p^{+}}_{p(x)q(x)},|u|^{p^{-}}_{p(x)q(x)})\leq||u|^{p(x)}|_{q(x)}\leq \max(|u|^{p^{-}}_{p(x)q(x)}, |u|^{p^{+}}_{p(x)q(x)}).$$
\end{prop}

  \begin{defin}\label{de1}
  Assume that spaces $E$, $F$ are Banach spaces, we define the norm on the space $X := E\cap F$
as $\|u\|_X = \|u\|_E + \|u\|_F$ .
\end{defin}

In order to discuss problems $\eqref{p1}$ and $\eqref{1}$, we need some properties of the space $X := W^{1,p(x)}_{0}(\Omega)\cap W^{2,p(x)}(\Omega)$.
From Definition \ref{de1}, we know that for any $u\in X$ we have $\|u\| = \|u\|_{1,p(x)}+\|u\|_{2,p(x)}$, thus  $\|u\| =|u|_{p(x)}+|\nabla u|_{p(x)}+\displaystyle\sum_{|\alpha|=2}|D^{\alpha}u|_{p(x)}.$
In Zang and Fu \cite{ZF}, the equivalence of the norms was proved, and it was even proved that the norm
$|\Delta u|_{p(x)}$ is equivalent to the norm $\|u\| $ (see \cite[Theorem 4.4]{ZF}).
 Note that $(X, \|.\|)$ is  a separable and
reflexive Banach space.

Let $$\|u\|_a:=\inf\left\{\mu>0:\displaystyle\int_{\Omega}\left(|\frac{\Delta u}{\mu}|^{p(x)}+a(x)|\frac{u}{\mu}|^{p(x)}\right)dx\leq 1 \right\}\quad\mbox{for}\ \ u\in X.$$
Since $a\in L^{\infty}(\Omega)$ and ${\rm essinf}_{x\in \Omega} a>0$, we deduce that  $\|u\|_a$ is equivalent to the norms $\|u\|$ and $|\Delta u|_{p(x)}$ in $X$. In our paper, we will use the norm $\|u\|_a$
and the modular is defined as
$\rho_{p(x)} : X \rightarrow \mathbb{R}$ by $$\rho_{p(x)}(u)=\int_{\Omega}|\Delta u|^{p(x)}+a(x)|u|^{p(x)}dx,$$ which satisfies the same properties as Proposition \ref{1}. Accordingly, we have the following property.

\begin{prop}\label{am}
For all $u\in L^{p(x)}(\Omega)$, we have
\begin{enumerate}
                       \item $\|u\|_{a}<1\ (resp. =1, >1) \Leftrightarrow \rho_{p(x)}(u)<1\ (resp. =1, >1).$
                       \item $\min(\|u\|_{a}^{p^{-}},\|u\|_{a}^{p^{+}})\leq\rho_{p(x)}(u)\leq \max(\|u\|_{a}^{p^{-}},\|u\|_{a}^{p^{+}}).$
                       \item $\|u_n\|_a \rightarrow 0\ (\mbox{respectively},\rightarrow\infty)\Leftrightarrow \rho_{p(x)}(u_n)\rightarrow 0\ (\mbox{respectively},\rightarrow\infty).$
                     \end{enumerate}
\end{prop}
\begin{prop}\label{111}
Let $L(u)=\int_{\Omega}\frac{1}{p(x)}(|\Delta u|^{p(x)}dx+a(x)|u|^{p(x)})dx$, then
\begin{enumerate}
                       \item $L:X\rightarrow\mathbb{R}$ is sequentially weakly lower semi continuous, $L\in C^1(X,\mathbb{R})$.
                       \item  The mapping $L':X\rightarrow X^*$ is a strictly monotone, bounded homeomorphism and is of type $(S_+)$, that is, if $u_n\rightharpoonup u$ and $\displaystyle\limsup_{n\rightarrow +\infty} L'(u_n)(u_n-u)\leq 0$, then $u_n\rightarrow u$.

                     \end{enumerate}
\end{prop}

We recall that the critical Sobolev exponent is defined as follows:
 $$
\left\{
\begin{array}{l}
p^*(x)=\displaystyle\frac{Np(x)}{N-2p(x)},\quad p(x)<\frac{N}{2},\\
p^*(x)=\displaystyle+\infty, \hspace{0.8cm}\quad\quad p(x)\geq \frac{N}{2}. \\
\end{array}
\right.
$$

We point out that if
 $q \in C^+(\overline{\Omega})$ and $q(x) < p^*(x)$ for all $x\in\overline\Omega$, then $X$ is
continuously and compactly embedded in $L^{q(x)}(\Omega)$.


The  Lebesgue and Sobolev
spaces with variable exponents coincide with the usual Lebesgue and Sobolev spaces provided that $p$ is constant. According to \cite[pp. 8-9]{radrep}, the function spaces $L^{p(x)}$ and $W^{1,p(x)}$ have some non-usual properties,
such as:

(i) If $p\geq 1$ is a real number, then the following co-area formula 
$$\int_\Omega |u(x)|^pdx=p\int_0^\infty t^{p-1}\,|\{x\in\Omega ;\ |u(x)|>t\}|\,dt$$
has  no analogue in the framework of variable exponents (namely, if $p:\overline\Omega\rightarrow [1,\infty)$ is a nonconstant smooth function).

(ii) Spaces $L^{p(x)}$ do {\it not} satisfy the {\it mean continuity property}. More exactly, if $p$ is nonconstant and continuous in an open ball $B$, then there is some $u\in L^{p(x)}(B)$ such that $u(x+h)\not\in L^{p(x)}(B)$ for every $h\in{\mathbb R}^N$ with arbitrary small norm.

(iii) Function spaces with variable exponent
 are {\it never} invariant with respect to translations.  The convolution is also limited. For instance,  the classical Young inequality
$$| f*g|_{p(x)}\leq C\, | f|_{p(x)}\, \| g\|_{L^1}$$
remains true if and only if
$p$ is constant.

\section{Auxiliary properties and main result for problem \eqref{p1}}
Throughout this section, the letters $ c, c_i, i=1,2,....$ denote positive constants which may change from line to line.
Let $s'_{1}(x)$ denote the conjugate exponent of the
function $s_1(x)$ and set $r_1(x):=\frac{s_1(x) q(x)}{s_1(x)-q(x)}$.
Then we have the following embedding property.

\begin{rem}\label{r1}
Assume that assumption $(H_2)$ is fulfilled, then $\max(r_1(x),s{'}_{1}(x)q(x))<p^{*}(x)$, for all $x\in \overline{\Omega}$, consequently
the embeddings $X\hookrightarrow L^{s'_{1}(x)
q(x)}(\Omega)$ and $X\hookrightarrow
L^{r_1(x)}(\Omega)$ are compact and continuous.
\end{rem}

\begin{defin}\label{d1}
We say that $\lambda\in \mathbb{R}$ is an eigenvalue of problem
\eqref{p1}, if there exists  $u\in X\setminus\{0\}$  such that $\Delta u=0$ on $\partial\Omega$ and
$$M\Big(\int_{\Omega}\frac{1}{p(x)}(|\Delta u|^{p(x)}dx+a|u|^{p(x)}dx)\Big)\ \int_{\Omega}(|\Delta u|^{p(x)-2}\Delta u \Delta v+a|u|^{p(x)-2}uv) dx=\lambda\int_{\Omega} V_1|u|^{q(x)-2}u v dx,$$
for any $v\in X$. If $\lambda$
is an eigenvalue of problem $(\ref{p1})$, then the corresponding
$u\in X\setminus\{0\}$ is a weak solution of problem
$(\ref{p1})$.
\end{defin}

\smallskip
The first main result of this part is the following.
\begin{teo}\label{thm1}
Assume that hypotheses $(H_0)$, $(H_1)$, $(H_2)$ are fulfilled. Then there exists
$\lambda^{*}>0$, such that any $\lambda\in (0,\lambda^{*})$ is an
eigenvalue of problem \eqref{p1}.
\end{teo}

 In order to describe the variational framework associated to \eqref{p1}, we define the functionals $\Phi,\, J:X\to \mathbb{R}$ defined as follows:
$$\Phi(u)=\widehat{M}\Big(\int_{\Omega}\frac{1}{p(x)}(|\Delta u|^{p(x)}+a(x)|u|^{p(x)})dx\Big)\  \ \mbox{and} \ J(u)=\int_{\Omega}\frac{V_1(x)}{q(x)}|u|^{q(x)}dx.$$

By Proposition \ref{norm1} and Remark \ref{r1}, $J$ is well defined and for all $u\in X$
$$\displaystyle{|J(u)| \leq \frac{1}{q^{-}} |V_1|_{s_1(x)}||u|^{q(x)}|_{s'_{1}(x)}}\leq \left\{
\begin{array}{l}
\frac{1}{q^{-}} |V_1|_{s_1(x)}|u|^{q^{-}}_{s'_{1}(x)q(x)},\quad \mbox{if}\ |u|_{s'_{1}(x)q(x)}\leq 1,\\
\frac{1}{q^{-}} |V_1|_{s_1(x)}|u|^{q^{+}}_{s'{1}(x)q(x)}, \quad\mbox{if}\  |u|_{s'_{1}(x)q(x)}>1 . \\
\end{array}
\right.$$

The Euler-Lagrange functional corresponding to problem
\eqref{p1} is $\Psi_{\lambda}:X\to \mathbb{R}$ and is defined by
$$
\Psi_\lambda(u):=\Phi(u)-\lambda J(u).
$$

\subsection{Proof of Theorem \ref{thm1}}
We start with the following auxiliary property.

\begin{prop}\label{pr00}
Assume that hypotheses $(H_0)$, $(H_1)$ and $(H_2)$ are fulfilled. Then $\Psi_\lambda\in C^{1}(X,\mathbb{R})$ is weakly lower semi-continuous
and $u\in X$ is a critical point of $\Psi_\lambda$ if and only if
$u$ is a weak solution of problem \eqref{p1}.
\end{prop}
\begin{proof}
To show that $\Psi_\lambda\in C^{1}(X,\mathbb{R})$, we establish that for
all $\varphi\in X$,
$$\lim_{{t\to 0^{+}}}{\frac{\Psi_\lambda(u+t\varphi)-\Psi_\lambda(u)}{t}}=\langle d\Psi_\lambda(u), \varphi\rangle,$$
and $d\Psi_\lambda:X\rightarrow X^{*}$ continuous, where we denote
by $X^{*}$ the dual space of $X$.

For all $\varphi\in X$ we have
\begin{eqnarray*}
\lim_{{t\to 0^{+}}}{\frac{J(u+t\varphi)-J(u)}{t}}
&=&\frac{d}{dt}J(u+t\varphi)|_{t=0}
=\frac{d}{dt}\int_{\Omega}\frac{V_1(x)}{q(x)}|u+t\varphi|^{q(x)}dx|_{t=0}\\
&=&\int_{\Omega}\frac{\partial}{\partial
t}\left(\frac{V_1(x)}{q(x)}|u+t\varphi|^{q(x)}\right)|_{t=0}dx\\
&=&\int_{\Omega}V_1(x)|u+t\varphi|^{q(x)-1}sgn(u+t\varphi)\varphi|_{t=0}dx\\
&=&\int_{\Omega}V_1(x)|u+t\varphi|^{q(x)-2}(u+t\varphi)\varphi|_{t=0}dx
=\int_{\Omega}V_1(x)|u|^{q(x)-2}u\varphi dx\\
&=&\langle d J(u), \varphi\rangle.
\end{eqnarray*}
The differentiation under the integral is allowed, since  for all $|t|<1$ we have:
$$|V_1(x)|u+t\varphi|^{q(x)-2}(u+t\varphi)\varphi|\leq |V_1(x)|(|u|+|\varphi|)^{q(x)-1}|\varphi|\in L^{1}(\Omega)
.$$
Since $u,\varphi\in X$ we have
$$|u|,|\varphi|\in X\hookrightarrow
L^{q(x)}(\Omega) \ \mbox{and}\ \ |\varphi|\in X\hookrightarrow
L^{r_1(x)}(\Omega).$$ Due to the fact that $V_1\in
L^{s_1(x)}(\Omega)$, the conclusion is an immediate consequence of
the generalized H\"older inequality (\ref{Hol1}).

Next, we show that for all $u\in X$,  $dJ(u)$ is in $
X^{*}$. We first observe that $dJ(u)$ is
linear.
Since there is a continuous embedding $X\hookrightarrow
L^{r_1(x)}(\Omega)$, we have
\begin{equation}\label{4}
\begin{gathered}
 $$|v|_{{r_1(x)}}\leq c \|v\|_a,\quad \mbox{for all } v\in X.$$
\end{gathered}
\end{equation}
 Using inequalities \eqref{Hol1} and \eqref{4} we
obtain
\begin{eqnarray*}
|\langle
dJ(u),\varphi\rangle|&=&|\int_{\Omega}V_1(x)|u|^{q(x)-2}u\varphi dx|\\
&\leq& \int_{\Omega}|V_1(x)||u|^{q(x)-1}|\varphi| dx\\ &\leq&
|V_1|_{s_1(x)}||u|^{q(x)-1}|_{\frac{q(x)}{q(x)-1}}|\varphi|_{r_1(x)}\\
&\leq& c |V_1|_{s_1(x)}||u|^{q(x)-1}|_{\frac{q(x)}{q(x)-1}} \|\varphi\|_a.
\end{eqnarray*}
Thus, there exists $c_1:= c|V_1(x)|
||u|^{q(x)-1}|_{\frac{q(x)}{q(x)-1}}>0$ such that $$|\langle
dJ(u),\varphi\rangle|\leq c_1 \|\varphi\|_a.$$ Using the linearity of
$dJ(u)$ and the above inequality we deduce that $dJ(u)\in
X^{*}$. For the Fr\'echet differentiability we
need the following auxiliary property.

\begin{lemma}[See \cite{KM}]\label{l0}
The map $u\in L^{q(x)}(\Omega)\mapsto |u|^{q(x)-2}u\in
L^{\frac{q(x)}{q(x)-1}}(\Omega)$ is continuous.
\end{lemma}
We conclude that $J$ is Fr\'echet differentiable.

The functional $\Phi$ is well defined, is continuously G\^ateaux differentiable and its G\^ateaux derivative is given by
$$\langle d\Phi(u),v\rangle=M\Big(\int_{\Omega}\frac{1}{p(x)}(|\Delta u|^{p(x)}dx+a(x)|u|^{p(x)}dx)\Big)\ \int_{\Omega}(|\Delta u|^{p(x)-2}\Delta u \Delta v+a(x)|u|^{p(x)-2}uv) dx,$$
for all $u,v\in X$.

We deduce that $\Psi_{\lambda}\in C^{1}(X,\RR)$ because $\Phi,\,J
\in C^{1}(X,\RR)$. Moreover
\begin{eqnarray*}
\langle d\Psi_{\lambda}(u),v\rangle=M\Big(\int_{\Omega}\frac{1}{p(x)}(|\Delta u|^{p(x)}dx+a(x)|u|^{p(x)}dx)\Big)\ \int_{\Omega}(|\Delta u|^{p(x)-2}\Delta u \Delta v+a(x)|u|^{p(x)-2}uv) dx\\
-\lambda\int_{\Omega}V_1(x)|u|^{q(x)-2}u v dx
\end{eqnarray*}
 for all $u,v \in X$.

Let $u$ be a critical point of $\Psi_{\lambda}$. Then we have
$d\Psi_{\lambda}(u)=0_{X^{*}}$, that is,
$$\langle d\Psi_{\lambda}(u),v\rangle=0,\quad \mbox{ for all } v\in X.$$
Therefore
$$\begin{array}{ll}
&\di M\Big(\int_{\Omega}\frac{1}{p(x)}(|\Delta u|^{p(x)}dx+a(x)|u|^{p(x)}dx)\Big)\, \int_{\Omega}(|\Delta u|^{p(x)-2}\Delta u \Delta v+a(x)|u|^{p(x)-2}uv) dx=\\
&\di\lambda\int_{\Omega}V_1(x) |u|^{q(x)-2}u v dx,\end{array}$$
for all $v\in X$.
It follows that $u$ is a weak solution of problem \eqref{p1}.

Now we assume that $u$ is a weak solution of  \eqref{p1}.
By Definition \ref{d1} we deduce that $\langle d\Psi_{\lambda}(u),v\rangle=0$, for all $v\in X.$ We
obtain $d\Psi_{\lambda}(u)=0_{X^{*}}$, hence $u$ is a critical point
of $\Psi_{\lambda}$. This completes the proof of Proposition
\ref{pr00}.
\end{proof}

The following property shows the existence of a {\it mountain} for $\Psi_\lambda$ near the origin.
\begin{lemma}\label{l5}
Suppose that the  hypotheses of Theorem \ref{thm1} are fulfilled. Then for all
$\rho \in (0,1)$, there exist $\lambda^*>0$ and $b>0$ such that for
all $u\in X$ with $\|u\|_a=\rho$
$$\Psi_{\lambda}(u)\geq b>0 \quad \quad \mbox{for all} \quad \quad \lambda \in (0,\lambda^*).$$
\end{lemma}

\begin{proof}
Since the embedding $X\hookrightarrow L^{s'_{1}(x)q(x)}(\Omega)$ is
continuous, we have
\begin{eqnarray}\label{ineg1}
|u|_{s'_{1}(x)q(x)}\leq c_2\|u\|_a,\ \mbox{for all}\ u\in X.
\end{eqnarray}
Let us assume
that $\|u\|_a<\min(1,1/c_2)$, where $c_2$ is the positive constant of inequality (\ref{ineg1}). It follows that $|u|_{s'_{1}(x)q(x)}<1$. Moreover, by hypothesis $(H_0)$, we have $\widehat{M}(t)\geq m_0 t$. Consequently, by combining H\"older's inequality
(\ref{7}), Proposition \ref{am} and inequality (\ref{ineg1}), we deduce that for all $u \in X$
with $\|u\|_a=\rho $,
\begin{eqnarray*}
\Psi_{\lambda}(u)&=&\widehat{M}\Big(\int_{\Omega}\frac{1}{p(x)}(|\Delta
u|^{p(x)}+a(x)|u|^{p(x)})dx\Big)- \frac{\lambda}{q^{-}}{\int_{\Omega}V_1(x)|
u|^{q(x)}dx}\\
&\geq&\frac{m_0}{p^{+}}\|u\|_{a}^{p^{+}}-\frac{\lambda }{q^{-}} |V_1|_{s_1(x)}||u|^{q(x)}|_{s'(x)}\\
&\geq&\frac{m_0}{p^{+}}\|u\|_{a}^{p^{+}}-\frac{\lambda }{q^{-}} |V_1|_{s_1(x)}|u|^{q^{-}}_{s'(x)q(x)}\\
&\geq&\frac{m_0}{p^{+}}\|u\|_{a}^{p^{+}}-\frac{\lambda }{q^{-}} |V_1|_{s_1(x)}c_{2}^{q^{-}}\|u\|_{a}^{q^{-}}\\
&=&\frac{m_0}{p^{+}}\rho^{p^{+}}-\frac{\lambda }{q^{-}}
c_{2}^{q^{-}}|V_1|_{s_1(x)}\rho^{q^{-}}=\rho^{q^-}
(\frac{m_0}{p^{+}}\rho^{p^{+}-q^{-}}-\frac{\lambda }{q^{-}} c_{2}^{q^{-}} |V_1|_{s_1(x)}).\\
\end{eqnarray*}
By the above inequality, we remark that if we define
\begin{eqnarray}\label{L16}
\displaystyle \lambda^*=\frac
{m_0\rho^{p^{+}-q^{-}}}{2p^+}\frac{q^{-}}{c_{2}^{q^-}|V_1|_{s_1(x)}},
\end{eqnarray}
then for any $\lambda \in (0,\lambda^*)$ and $u \in X$ with
$\|u\|_a=\rho$ there exists $ b>0$ such that
$$\Psi_{\lambda}(u)\geq b>0.$$
The proof of Lemma \ref{l5} is complete.
\end{proof}

The following result asserts the existence of a {\it valley} for $\Psi_\lambda$ near the origin.
\begin{lemma}\label{l6}
There exists $\phi \in X\setminus\{0\}$ such that $\phi \geq0$ and
$\Psi_{\lambda}(t\phi)<0$, for $t>0$ small enough.
\end{lemma}

\begin{proof}
By hypothesis $(H_1)$, there exists $t_0>0$ such that
    for all $t>t_0$, we have $$\widehat{M}(t)\leq\frac{\widehat{M}(t_0)}{t_0^{\frac{1}{1-\theta}}}t^{\frac{1}{1-\theta}}=c_3 t^{\frac{1}{1-\theta}}.$$
    Moreover, by $(H_2)$, we have $q(x)<p(x)$, for all  $x\in
\overline{\Omega}_0$.

In the sequel, we denote
$$q_{0}^{-}:=\inf_{\Omega_0}q(x)\quad \mbox{and}\quad
p_{0}^{-}:=\inf_{\Omega_0}p(x).$$
 Let $\epsilon_0$ be such that
$q_{0}^{-}+\epsilon_0<p_{0}^{-}$. Since $q\in
C(\overline{\Omega}_0)$, there exists an open set
$\Omega_1\subset\Omega_0$ such that $|q(x)-q_{0}^{-}|<\epsilon_0,\
\mbox{for all}\ x\in \Omega_1$. It follows that
$q(x)\leq q_{0}^{-}+\epsilon_0<p_{0}^{-},\ \mbox{for all}\ x\in \Omega_1$.

  Let $\phi \in C_{0}^{\infty}(\Omega)$ be such that ${\rm supp}\,(\phi)\subset \Omega_1\subset\Omega_0$, $\phi=1$ in a subset $\Omega'_{1}\subset {\rm supp}\,(\phi)$, and $0\leq\phi\leq 1$ in $\Omega_{1}$. It follows that
\begin{eqnarray*}
\Psi_{\lambda}(t\phi)&=&\widehat{M}\Big(\int_{\Omega}\frac{1}{p(x)}(|\Delta
(t\phi)|^{p(x)}+a(x)|t\phi|^{p(x)})dx\Big)-\lambda\int_{\Omega}\frac{V_1(x)}{q(x)}|
t\phi|^{q(x)}dx\\
&\leq&c_3\Big(\int_{\Omega_{0}}\frac{t^{p(x)}}{p(x)}\big(|\Delta\phi|^{p(x)}+a(x)|\phi|^{p(x)})dx\Big)^{\frac{1}{1-\theta}}-\lambda\int_{\Omega_1}\frac{V_1(x)}{q(x)}t^{q(x)}V(x)|\phi|^{q(x)}dx\\
&\leq&\displaystyle
\frac{t^{\frac{p_{0}^-}{1-\theta}}}{(p_{0}^-)^{\frac{1}{1-\theta}}}\Big(\int_{\Omega_0}(|\Delta
\phi|^{p(x)}+a(x)|\phi|^{p(x)})dx\Big)^{\frac{1}{1-\theta}}-\frac{\lambda t^{q_{0}^{-}+\epsilon_0}}{q_{0}^{+}}\int_{\Omega_1}V_1(x)|\phi|^{q(x)}dx\\
&\leq&\displaystyle c_4 t^{p_{0}^{-}}\Big(\int_{\Omega_0}(|\Delta
\phi|^{p(x)}+a(x)|\phi|^{p(x)})dx\Big)^{\frac{1}{1-\theta}}-\frac{\lambda t^{q_{0}^{-}+\epsilon_0}}{q_{0}^{+}}\int_{\Omega_1}V_1(x)|\phi|^{q(x)}dx.\\
\end{eqnarray*}
Therefore
$$\displaystyle \Psi_{\lambda}(t\phi)<0$$
for $\displaystyle t<\delta^{1/(p_{0}^{-}-q_{0}^{-}-\epsilon_0)}$
with
$$\displaystyle 0<\delta<\mbox{min}\left\{1,\frac{\frac{\lambda}{c_4 q_{0}^+}\int_{\Omega_1}V_1(x)|\phi|^{q(x)}dx}{\Big(\int_{\Omega_0}(|\Delta
\phi|^{p(x)}+a(x)|\phi|^{p(x)})dx\Big)^{\frac{1}{1-\theta}}}\right\}.$$ Finally, we point out that
$$\int_{\Omega_0}(|\Delta
\phi|^{p(x)}+a(x)|\phi|^{p(x)})dx>0.$$
Indeed, supposing the contrary we have
$\int_{\Omega}(|\Delta\phi|^{p(x)}+a(x)|u|^{p(x)})dx=0$.
By Proposition \ref{am}, we deduce that $\|\phi\|_a=0$ and consequently $\phi=0$ in $\Omega$, a contradiction.
The proof of Lemma \ref{l6} is complete.
\end{proof}

{\it\bf{ Proof of Theorem \ref{thm1} completed.}} Let $\lambda^*>0$ be defined as in
\eqref{L16} and $\lambda \in (0,\lambda^*)$. By Lemma \ref{l5} it
follows that on the boundary of the ball centered at the origin and
of radius $\rho$ in $X$, denoted by $B_{\rho}(0)$, we have
\begin{equation}\label{L17}
\inf_{\partial B_{\rho}(0)}\Psi_{\lambda}>0.
\end{equation}
On the other hand, by Lemma \ref{l6}, there exists $\phi \in X$ such
that $\Psi_{\lambda}(t\phi)<0$ for all $t>0$ small enough. Moreover,
by hypothesis $(H_0)$, H\"older's inequality
(\ref{7}), Proposition \ref{am} and inequality (\ref{ineg1}), we deduce that for any $u
\in B_{\rho}(0)$ we have
\begin{equation*}
\Psi_{\lambda}(u)\geq
\displaystyle\frac{m_0}{p^{+}}\|u\|_{a}^{p^{+}}-\frac{\lambda}{q^{-}}c_{2}^{q-}|V_1|_{s_1(x)}\|u\|_{a}^{q^{-}}{{\bf
.}}
\end{equation*}
 It follows that
$$-\infty<\underline{c}:=\inf_{\overline{B_{\rho}(0)}}\Psi_{\lambda}<0.$$
Let $ 0<\epsilon<\inf_{\partial
B_{\rho}(0)}\Psi_{\lambda}-\inf_{B_{\rho}(0)}\Psi_{\lambda}$. Due to
the above information, the functional $\Psi_{\lambda}:\overline{B_{\rho}(0)}\longrightarrow \mathbb{R}$ is
lower bounded on $\overline{B_{\rho}(0)}$ and $\Psi_{\lambda}\in
C^{1}(\overline{B_{\rho}(0)},\RR)$. Thus, by Ekeland's variational
principle \cite{EK}, there exists $u_\epsilon \in \overline{B_{\rho}(0)}$ such
that
\begin{equation*}
\left\{
\begin{array}{lll}
\displaystyle \underline{c} \leq \Psi_{\lambda}(u_\epsilon)  \leq \underline{c}+\epsilon \\
\displaystyle 0<
\Psi_{\lambda}(u)-\Psi_{\lambda}(u_\epsilon)+\epsilon \cdot
\parallel u-u_\epsilon \parallel_a, \quad u\neq u_\epsilon.
\end{array}
\right.
\end{equation*}
Since
$$\Psi_{\lambda}(u_\epsilon)\leq\inf_{\overline{B_{\rho}(0)}}\Psi_{\lambda}+\epsilon \leq \inf_{B_{\rho}(0)}\Psi_{\lambda}
+\epsilon <\inf_{\partial B_{\rho}(0)}\Psi_{\lambda},$$ we deduce
that $u_\epsilon \in {B_{\rho}(0)}.$ Now, we define
$I_{\lambda}:\overline{B_{\rho}(0)}\longrightarrow\mathbb{R}$ by
$I_{\lambda}(u)= \Psi_{\lambda}(u)+\epsilon \cdot \parallel
u-u_\epsilon \parallel_a.$ It is clear that $u_\epsilon$ is a minimum
point of $I_{\lambda}$ and thus
$$\frac{I_{\lambda}(u_\epsilon+t\cdot v)-I_{\lambda}(u_\epsilon)}{t}\geq 0$$
for small $t>0$ and any $v \in B_{1}(0)$. The above relation yields
$$\frac{\Psi_{\lambda}(u_\epsilon+t\cdot v)-\Psi_{\lambda}(u_\epsilon)}{t} + \epsilon\cdot\parallel v\parallel_a\geq 0$$
Letting $t\rightarrow 0$ it follows that
$<d\Psi_{\lambda}(u_\epsilon),v>+\epsilon\cdot\parallel v \parallel_a
\geq0 $
and we infer that $\parallel d\Psi_{\lambda}(u_\epsilon)\parallel_a \leq\epsilon $.\\
We deduce that there exists a sequence $\{w_n\}\subset B_{\rho}(0)$
such that
\begin{equation}\label{L170}
  \Psi_{\lambda}(w_n) \longrightarrow \underline{c}\quad \mbox{and} \quad  d\Psi_{\lambda}(w_n)\longrightarrow 0_{X^*}.
\end{equation}

The sequence $\{w_n\}$ is bounded in $X$. Thus, there exists $w$
in $X$ such that, up to a subsequence, $\{w_n\}$ converges weakly to
$w$ in $X$. Since $r_1(x)<p^*(x)$ for all $x\in
\overline{\Omega}$ we deduce that there exists a compact embedding
$E\hookrightarrow L^{r_1(x)}(\Omega)$ and consequently $\{w_n\}$
converges strongly in $L^{r_1(x)}(\Omega)$. In order to have strong
convergence, we need the following auxiliary result.

\begin{prop}\label{lim} We have
$$ \displaystyle\lim_{n\to
\infty} \int_{\Omega}V_1(x)|w_{n}|^{q(x)-2}w_{n}(w_{n}-w)dx=0.$$
\end{prop}

\begin{proof}
Using H\"older's inequality (\ref{7}) we have
\begin{eqnarray*}
\displaystyle\int_{\Omega}V_1(x)|w_{n}|^{q(x)-2}w_{n}(w_{n}-w)dx &\leq& |V_1|_{s_1(x)}||w_n|^{q(x)-2}w_n(w_n-w)|_{s'_{1}(x)}\\
 &\leq& |V_1|_{s_1(x)}|||w_n|^{q(x)-2}w_n|_{\frac{q(x)}{q(x)-1}}|w_n-w|_{r_1(x)}.
\end{eqnarray*}
If $||w_n|^{q(x)-2}w_n|_{\frac{q(x)}{q(x)-1}}>1$, by Proposition
\ref{norm1}, we have\\
$$||w_n|^{q(x)-2}w_n|_{\frac{q(x)}{q(x)-1}}\leq
|w_n|^{q^{+}}_{q(x)}.$$ Using now the compact embedding $X\hookrightarrow
L^{q(x)}(\Omega)$, we conclude the proof.
\end{proof}

Since $d\Psi_{\lambda}(w_n)\to  0$ and $\{w_n\}$ is bounded in $X$ we
have
\begin{eqnarray*}
|\langle d\Psi_{\lambda}(w_n),w_n-w\rangle|&\leq&
|\langle d\Psi_{\lambda}(w_n),w_n\rangle|+|\langle d\Psi_{\lambda}(w_n),w\rangle|\\
&\leq&\|d\Psi_{\lambda}(w_n)\|_a\| w_n\|_a+ \| d\Psi_{\lambda}(w_n)\|_a \|
w\|_a.
\end{eqnarray*}
Moreover, by Proposition \ref{lim}, we have
$$
\lim_{n\to \infty}\langle d\Psi_{\lambda}(w_n),w_n-w\rangle=0.
$$
Therefore
\begin{eqnarray*}
\displaystyle\lim_{n\to \infty}M\Big(\int_{\Omega}\frac{1}{p(x)}(|\Delta w_n|^{p(x)}+a(x)|w_n|^{p(x)})dx\Big)\ \int_{\Omega}\Big(|\Delta w_n|^{p(x)-2}\Delta w_n (\Delta w_n-\Delta w)+\\a(x)|w_n|^{p(x)-2}w_n(w_n-w)\Big)dx=0.
\end{eqnarray*}
Combining hypothesis $(H_0)$ and Proposition \ref{111},  we deduce that
$\{w_n\}$ converges strongly to $w$ in $X$.
Since $\Psi_{\lambda}\in C^1(X,\mathbb{R})$, we conclude that
\begin{equation}\label{191}
 d\Psi_{\lambda}(w_n)\rightarrow d\Psi_{\lambda}(w),\ \mbox{as}\ n\rightarrow \infty.
\end{equation}

Now, relations \eqref{L170} and \eqref{191} yield
\begin{equation}\label{critica}
\Psi_{\lambda}(w)=\underline{c}<0\ \mbox{ and}\ \ d\Psi_{\lambda}(w)=0.
\end{equation}

In order to show that $w$ is a
solution of problem $(\ref{p1})$, it remains to show that $\Delta w=0$ on $\partial\Omega$.
Due to relation \eqref{critica}, $w\in X\setminus\{0\}$ is a critical point of $\Psi_\lambda$, so
\begin{equation}\label{star1}M(t)\int_\Omega |\Delta w|^{p(x)-2}\Delta w\Delta vdx=\int_\Omega m(x)vdx\quad\mbox{for all}\ v\in X,
\end{equation}
where
$$m(x)=\lambda V_1(x)|w|^{q(x)-2}w-M(t)a(x)|w|^{p(x)-2}w.$$
Relation (\ref{star1}) implies that
\begin{equation}\label{1star1}M(t)\int_\Omega |\Delta w|^{p(x)-2}\Delta w\Delta vdx=\int_\Omega m(x)vdx\quad\mbox{for all}\ v\in C^\infty_0(\Omega).\end{equation}

Let $\zeta$ be the unique solution of the problem
$$\left\{\begin{array}{lll}
&\Delta\zeta =m(x)\quad&\mbox{in}\ \Omega\\
&\zeta =0\quad&\mbox{on}\ \partial\Omega.\end{array}\right.
$$
Relation (\ref{1star1}) yields
$$M(t)\int_\Omega |\Delta w|^{p(x)-2}\Delta w\Delta vdx=\int_\Omega (\Delta\zeta)vdx\quad\mbox{for all}\ v\in C^\infty_0(\Omega).$$Using the Green formula we have
$$\int_\Omega (\Delta\zeta)vdx=\int_\Omega \zeta\Delta vdx.$$
Therefore
\begin{equation}\label{2star1}
M(t)\int_\Omega |\Delta w|^{p(x)-2}\Delta w\Delta vdx=\int_\Omega\zeta \Delta vdx\quad\mbox{for all}\ v\in C^\infty_0(\Omega).\end{equation}

On the other hand, for all $\tilde{w}\in C^\infty_0(\Omega)$ there exists a unique $v\in C^\infty_0(\Omega)$ such that $\Delta v=\tilde{w}$ in $\Omega$. Thus, relation (\ref{2star1}) can be rewritten as
$$\int_\Omega (M(t)|\Delta w|^{p(x)-2}\Delta w-\zeta)\tilde{w}dx=0\quad\mbox{for all}\ \tilde{w}\in C^\infty_0(\Omega).$$
Applying the fundamental lemma of the calculus of variations, we deduce that
$$M(t)|\Delta w|^{p(x)-2}\Delta w-\zeta=0\quad\mbox{in}\ \Omega.$$
According to assertion $(H_0)$ and since $\zeta=0$ on $\partial\Omega$, we conclude that $\Delta w=0$ on $\partial\Omega$.
Thus, $w$ is a  nontrivial weak solution of problem
\eqref{p1} such that $\Delta w=0$. Since $\Psi_{\lambda}(|w|)=\Psi_{\lambda}(w)$
then problem \eqref{p1} has a non-negative solution. The proof is now complete.\quad$\square$

%{\sc \bf{Proof of Theorem \ref{thm}.}} Using condition $(H_0)$ and H$\ddot{o}$lder inequality (\ref{7}) for %$\|u\|_a>1$, we have
%\begin{eqnarray*}
%\Psi_{\lambda}(u)&=&\widehat{M}\Big(\int_{\Omega}\frac{1}{p(x)}(|\Delta
%u|^{p(x)}+a(x)|u|^{p(x)})dx\Big)- \frac{\lambda}{q^{-}}{\int_{\Omega}V(x)|
%u|^{q(x)}dx}\\
%&\geq&\frac{m_0}{p^{+}}\|u\|_{a}^{p^{-}}-\frac{\lambda }{q^{-}} |V|_{s(x)}||u|^{q(x)}|_{s'(x)}\\
%&\geq&\frac{m_0}{p^{+}}\|u\|_{a}^{p^{-}}-\frac{\lambda }{q^{-}} |V|_{s(x)}|u|^{q^{-}}_{s'(x)q(x)}\\
%&\geq&\frac{m_0}{p^{+}}\|u\|_{a}^{p^{-}}-\frac{\lambda }{q^{-}} |V|_{s(x)}C^{q^{-}}\|u\|_{a}^{q^{-}}\rightarrow %+\infty,\ \mbox{as} \ \|u\|_a\rightarrow +\infty.\\
%\end{eqnarray*}
%As a conclusion, since $\Psi_{\lambda}$ is weakly lower semi-continuous then it has a global minimizer which is %solution of problem (\ref{1}), moreover lemma \ref{l6} ensures that this solution is not trivial, which ends the %proof.\quad$\square$



\section{Main result for problem \eqref{1}}
In what follows assume we the following hypothesis:\\
$(H_3)$ $1<q(x)<\alpha(x)<p(x)< \frac{N}{2}<\min(s_1(x),s_2(x))$, $\mbox{for all }x\in \overline{\Omega}$, where $s_2\in C(\overline\Omega)$ and \\ $V_2\in L^{s_2(x)}(\Omega)$ such that $V_2\geq0$ in $\Omega$.

 Let
$s_{2}'(x)$ denote the conjugate exponent of the function
 $s_2(x)$ and
$r_2(x):=\frac{s_2(x) \alpha(x)}{s_2(x)-\alpha(x)}$. Then the following embedding properties hold.

\begin{rem}\label{r}
Under assumption $(H_3)$, we have $\displaystyle\max(r_2(x)s_2'(x)\alpha(x))<p^{*}(x),\ \mbox{for all}\ x\in
\overline{\Omega}$. Consequently, the embeddings
$X\hookrightarrow L^{s_2'(x)
\alpha(x)}(\Omega)$ and $X\hookrightarrow
L^{r_2(x)}(\Omega)$ are compact and continuous.
\end{rem}


\begin{defin}\label{d11}
We say that $\lambda\in \mathbb{R}$ is an eigenvalue of problem
\eqref{1}, if there exists  $u\in X\setminus\{0\}$  such that $\Delta u=0$ on $\partial\Omega$ and
$$M(t)\int_{\Omega}(|\Delta u|^{p(x)-2}\Delta u \Delta v+a(x)|u|^{p(x)-2}uv) dx=\lambda\int_{\Omega}(V_1(x)|u|^{q(x)-2}- V_2(x) |u|^{\alpha(x)-2})uvdx,$$
for any $v\in X$. If $\lambda$
is an eigenvalue of problem $(\ref{1})$, then the corresponding
$u\in X\setminus\{0\}$ is a weak solution of problem
$(\ref{1})$.
\end{defin}

\smallskip
The main result on this part of the paper is the following.
\begin{teo}\label{thm}
Assume that hypotheses $(H_0)$, $(H_1)$ and $(H_3)$ are fulfilled. Then there exists
$\lambda^{*}>0$, such that any $\lambda\in (0,\lambda^{*})$ is an
eigenvalue of problem \eqref{1}.
\end{teo}

 In order to describe the variational framework associated to \eqref{1}, we define the functional $\gamma :X\to \mathbb{R}$  as follows:
$$ \gamma(u)=\int_{\Omega}\frac{V_2(x)}{\alpha(x)}|u|^{\alpha(x)}dx.$$

By Proposition \ref{norm1} and Remark \ref{r}, $\gamma$ is well defined and for all $u\in X$
$$|\gamma(u)|\leq\frac{1}{\alpha^-}|V_2|_{s_2(x)}||u|^{\alpha(x)}|_{s_{2}'(x)}\leq\left\{
\begin{array}{l}
\frac{1}{\alpha^{-}} |V_2|_{s(x)}|u|^{\alpha^{-}}_{s_{2}'(x)\alpha(x)},\quad \mbox{if}\ |u|_{s_{2}'(x)\alpha(x)}\leq 1,\\
\frac{1}{\alpha^{-}} |V_2|_{s(x)}|u|^{\alpha^{+}}_{s_{2}'(x)\alpha(x)}, \quad\mbox{if}\  |u|_{s_{2}'(x)\alpha(x)}>1 . \\
\end{array}
\right.$$

The Euler-Lagrange functional corresponding to problem
\eqref{1}  is $\Psi_{\lambda}:X\to \mathbb{R}$ and is defined by
$$
\Psi_\lambda(u):=\Phi(u)-\lambda J(u)+\lambda \gamma(u).
$$
According to Proposition \ref{111}, one has $\Phi\in C^1(X,\mathbb{R})$. Moreover, under assumption $(H_3)$ and Proposition 2 in \cite{MK} one has $J, \gamma\in C^1(X,\mathbb{R})$, so $\Psi_\lambda\in C^1(X,\mathbb{R})$ and
\begin{eqnarray*}
\langle d\Psi_{\lambda}(u),v\rangle=M\Big(\int_{\Omega}\frac{1}{p(x)}(|\Delta u|^{p(x)}dx+a(x)|u|^{p(x)}dx)\Big)\ \int_{\Omega}(|\Delta u|^{p(x)-2}\Delta u \Delta v+a(x)|u|^{p(x)-2}uv) dx\\
-\lambda\int_{\Omega}V_1(x)|u|^{q(x)-2}u v dx+\lambda\int_{\Omega}V_2(x)|u|^{\alpha(x)-2}u v dx,\\
\end{eqnarray*}
 for all $u,v \in X$.






\subsection{Proof of Theorem \ref{thm}}
The following property shows the existence of a {\it mountain} for $\Psi_\lambda$ near the origin.
\begin{lemma}\label{l55}
Suppose that the  hypotheses of Theorem \ref{thm} are fulfilled. Then for all
$\rho \in (0,1)$, there exists $\lambda^*>0$ and $b>0$ such that for
all $u\in X$ with $\|u\|_a=\rho$
$$\Psi_{\lambda}(u)\geq b>0 \quad \quad \mbox{for all} \quad \quad \lambda \in (0,\lambda^*).$$
\end{lemma}

\begin{proof}
Let us assume
that $\|u\|_a<\min(1,1/c_2)$, where $c_2$ is the positive constant of inequality (\ref{ineg1}). It follows that $|u|_{s'_{1}(x)q(x)}<1$. Since $V_2\geq 0$ on $\Omega$, we use the same steps as the proof of Lemma \ref{l55} to deduce
 that for all $u \in X$
with $\|u\|_a=\rho $,
\begin{eqnarray*}
\Psi_{\lambda}(u)&\geq&\widehat{M}\Big(\int_{\Omega}\frac{1}{p(x)}(|\Delta
u|^{p(x)}+a(x)|u|^{p(x)})dx\Big)- \frac{\lambda}{q^{-}}{\int_{\Omega}V_1(x)|
u|^{q(x)}dx}+\frac{\lambda}{\alpha^{+}}{\int_{\Omega}V_2(x)|u|^{\alpha(x)} dx}\\
&\geq&\frac{m_0}{p^{+}}\|u\|_{a}^{p^{+}}-\frac{\lambda }{q^{-}} |V_1|_{s_1(x)}||u|^{q(x)}|_{s_{1}'(x)}\\
&=&\frac{m_0}{p^{+}}\rho^{p^{+}}-\frac{\lambda }{q^{-}}
c_{2}^{q^{-}}|V_1|_{s_1(x)}\rho^{q^{-}}=\rho^{q^-}
(\frac{m_0}{p^{+}}\rho^{p^{+}-q^{-}}-\frac{\lambda }{q^{-}} c_{2}^{q^{-}} |V_1|_{s_1(x)}).\\
\end{eqnarray*}
By the above inequality, we remark that if we define
\begin{eqnarray}\label{L166}
\displaystyle \lambda^*=\frac
{m_0\rho^{p^{+}-q^{-}}}{2p^+}\frac{q^{-}}{c_{2}^{q^-}|V_1|_{s_1(x)}},
\end{eqnarray}
then for any $\lambda \in (0,\lambda^*)$ and $u \in X$ with
$\|u\|_a=\rho$ there exists $ b>0$ such that
$$\Psi_{\lambda}(u)\geq b>0.$$
The proof of Lemma \ref{l55} is complete.
\end{proof}

The following result asserts the existence of a {\it valley} for $\Psi_\lambda$ near the origin.
\begin{lemma}\label{l66}
There exists $\phi \in X\setminus\{0\}$ such that $\phi \geq0$ and
$\Psi_{\lambda}(t \phi)<0$, for $t>0$ small enough.
\end{lemma}

\begin{proof}
By assumption $(H_3)$, one has $q(x)<\alpha(x)$, for all  $x\in
\overline{\Omega}_0$.
In the sequel, we denote
$$q_{0}^{-}:=\inf_{\Omega_0}q(x),\
p_{0}^{-}:=\inf_{\Omega_0}p(x) \quad\mbox{and}\quad \alpha_{0}^{-}:=\inf_{\Omega_0}\alpha(x).$$
 Let $\epsilon_0$ be such that
$q_{0}^{-}+\epsilon_0<\alpha_{0}^{-}$. Since $q\in
C(\overline{\Omega}_0)$, there exists an open set
$\Omega_1\subset\Omega_0$ such that $|q(x)-q_{0}^{-}|<\epsilon_0,\
\mbox{for all}\ x\in \Omega_1$. It follows that
$q(x)\leq q_{0}^{-}+\epsilon_0<\alpha_{0}^{-},\ \mbox{for all}\ x\in \Omega_1$.

  Let $\phi \in C_{0}^{\infty}(\Omega)$ be such that ${\rm supp}\,(\phi)\subset \Omega_1\subset\Omega_0$, $\phi=1$ in a subset $\Omega'_{1}\subset {\rm supp}\,(\phi)$, and $0\leq\phi\leq 1$ in $\Omega_{1}$. It follows that
\begin{eqnarray*}
\Psi_{\lambda}(t\phi)&=&\widehat{M}\Big(\int_{\Omega}\frac{1}{p(x)}(|\Delta
(t\phi)|^{p(x)}+a(x)|t\phi|^{p(x)})dx\Big)-\lambda\int_{\Omega}\frac{V_1(x)}{q(x)}|
t\phi|^{q(x)}dx\\
&+&\lambda\int_{\Omega}\frac{t^{\alpha(x)}}{\alpha(x)}V_2(x)|\phi|^{\alpha(x)} dx\\
&\leq&c_3\Big(\int_{\Omega_{0}}\frac{t^{p(x)}}{p(x)}\left(|\Delta\phi|^{p(x)}+
a(x)|\phi|^{p(x)})dx\right)^{\frac{1}{1-\theta}}-\lambda\int_{\Omega_1}\frac{V_1(x)}{q(x)}t^{q(x)}|\phi|^{q(x)}dx\\
&+&\lambda\int_{\Omega_0}\frac{t^{\alpha(x)}}{\alpha(x)}V_2(x)|\phi|^{\alpha(x)} dx\\
&\leq&\displaystyle
c_3\frac{t^{\frac{p_{0}^-}{1-\theta}}}{(p_{0}^-)^{\frac{1}{1-\theta}}}\Big(\int_{\Omega_0}(|\Delta
\phi|^{p(x)}+a(x)|\phi|^{p(x)})dx\Big)^{\frac{1}{1-\theta}}-\frac{\lambda t^{q_{0}^{-}+\epsilon_0}}{q_{0}^{+}}\int_{\Omega_1}V_1(x)|\phi|^{q(x)}dx\\
&+&\lambda\frac{t^{\alpha_{0}^{-}}}{\alpha_{0}^{-}}\int_{\Omega_0}V_2(x)|\phi|^{\alpha(x)} dx\\
&\leq&\displaystyle c_3\frac{ t^{p_{0}^{-}}}{(p_{0}^{-})^{\frac{1}{1-\theta}}}\Big(\int_{\Omega_0}(|\Delta
\phi|^{p(x)}+a(x)|\phi|^{p(x)})dx\Big)^{\frac{1}{1-\theta}}-\frac{\lambda t^{q_{0}^{-}+\epsilon_0}}{q_{0}^{+}}\int_{\Omega_1}V_1(x)|\phi|^{q(x)}dx\\
&+&\lambda\frac{t^{\alpha_{0}^{-}}}{\alpha_{0}^{-}}\int_{\Omega_0}V_2(x)|\phi|^{\alpha(x)} dx\\
&\leq&\displaystyle\frac{\max(c_3,1)t^{\alpha_{0}^{-}}}{\min(\alpha_{0}^{-},(p_{0}^{-})^{\frac{1}{1-\theta}})} \Big[\int_{\Omega_0}\Big(|\Delta
\phi|^{p(x)}+a(x)|\phi|^{p(x)})dx\Big)^{\frac{1}{1-\theta}}+\lambda\int_{\Omega_0}V_2(x)|\phi|^{\alpha(x)} dx\Big]\\
&-&\frac{\lambda t^{q_{0}^{-}+\epsilon_0}}{q_{0}^{+}}\int_{\Omega_1}V_1(x)|\phi|^{q(x)}dx\\
\end{eqnarray*}
Therefore
$$\displaystyle \Psi_{\lambda}(t\phi)<0$$
for $\displaystyle t<\delta^{1/(\alpha_{0}^{-}-q_{0}^{-}-\epsilon_0)}$
with
$$\displaystyle 0<\delta<\mbox{min}\left\{1,\frac{\frac{\lambda\min(\alpha_{0}^{-},(p_{0}^{-})^{\frac{1}{1-\theta}})}{\max(c_3,1) q_{0}^+}\int_{\Omega_1}V_1(x)|\phi|^{q(x)}dx}{\Big(\int_{\Omega_0}(|\Delta
\phi|^{p(x)}+a(x)|\phi|^{p(x)})dx\Big)^{\frac{1}{1-\theta}}+\int_{\Omega_0}V_2(x)|\phi|^{\alpha(x)} dx}\right\}.$$ Finally, we point out that
$$\Big(\int_{\Omega_0}(|\Delta
\phi|^{p(x)}+a(x)|\phi|^{p(x)})dx\Big)^{\frac{1}{1-\theta}}+\int_{\Omega_0}V_2(x)|\phi|^{\alpha(x)} dx>0.$$
Indeed, supposing the contrary we have
$\Big(\int_{\Omega}(|\Delta\phi|^{p(x)}+a(x)|u|^{p(x)}dx)\Big)^{\frac{1}{1-\theta}}+\int_{\Omega_0}V_2(x)|\phi|^{\alpha(x)} dx=0$.
By Proposition \ref{am}, we deduce that $\|\phi\|_a=0$ and consequently $\phi=0$ in $\Omega$, a contradiction.
The proof of Lemma \ref{l66} is complete.
\end{proof}

{\it {\bf Proof of Theorem \ref{thm} completed.}} Let $\lambda^*>0$ be defined as in
\eqref{L166} and $\lambda \in (0,\lambda^*)$. By Lemma \ref{l55} it
follows that on the boundary of the ball centered at the origin and
of radius $\rho$ in $X$, denoted by $B_{\rho}(0)$, we have
\begin{equation}\label{L177}
\inf_{\partial B_{\rho}(0)}\Psi_{\lambda}>0.
\end{equation}
On the other hand, by Lemma \ref{l66}, there exists $\phi \in X$ such
that $\Psi_{\lambda}(t\phi)<0$ for all $t>0$ small enough. Moreover,
by hypothesis $(H_0)$, H\"older's inequality
(\ref{7}), Proposition \ref{am} and inequality (\ref{ineg1}), we deduce that for any $u
\in B_{\rho}(0)$ we have
\begin{equation*}
\Psi_{\lambda}(u)\geq
\displaystyle\frac{m_0}{p^{+}}\|u\|_{a}^{p^{+}}-\frac{\lambda}{q^{-}}c_{2}^{q-}|V_1|_{s_1(x)}\|u\|_{a}^{q^{-}}{{\bf
.}}
\end{equation*}
 It follows that
$$-\infty<\underline{c}:=\inf_{\overline{B_{\rho}(0)}}\Psi_{\lambda}<0.$$
Let $ 0<\epsilon<\inf_{\partial
B_{\rho}(0)}\Psi_{\lambda}-\inf_{B_{\rho}(0)}\Psi_{\lambda}$. Due to
the above information, the functional $\Psi_{\lambda}:\overline{B_{\rho}(0)}\longrightarrow \mathbb{R}$ is
lower bounded on $\overline{B_{\rho}(0)}$ and $\Psi_{\lambda}\in
C^{1}(\overline{B_{\rho}(0)},\RR)$. Thus, by Ekeland's variational
principle, there exists $u_\epsilon \in \overline{B_{\rho}(0)}$ such
that
\begin{equation*}
\left\{
\begin{array}{lll}
\displaystyle \underline{c} \leq \Psi_{\lambda}(u_\epsilon)  \leq \underline{c}+\epsilon \\
\displaystyle 0<
\Psi_{\lambda}(u)-\Psi_{\lambda}(u_\epsilon)+\epsilon \cdot
\parallel u-u_\epsilon \parallel_a, \quad u\neq u_\epsilon.
\end{array}
\right.
\end{equation*}
Since
$$\Psi_{\lambda}(u_\epsilon)\leq\inf_{\overline{B_{\rho}(0)}}\Psi_{\lambda}+\epsilon \leq \inf_{B_{\rho}(0)}\Psi_{\lambda}
+\epsilon <\inf_{\partial B_{\rho}(0)}\Psi_{\lambda},$$ we deduce
that $u_\epsilon \in {B_{\rho}(0)}.$ Now, we define
$I_{\lambda}:\overline{B_{\rho}(0)}\longrightarrow\mathbb{R}$ by
$I_{\lambda}(u)= \Psi_{\lambda}(u)+\epsilon \cdot \parallel
u-u_\epsilon \parallel_a.$ It is clear that $u_\epsilon$ is a minimum
point of $I_{\lambda}$ and thus
$$\frac{I_{\lambda}(u_\epsilon+t\cdot v)-I_{\lambda}(u_\epsilon)}{t}\geq 0$$
for small $t>0$ and any $v \in B_{1}(0)$. The above relation yields
$$\frac{\Psi_{\lambda}(u_\epsilon+t\cdot v)-\Psi_{\lambda}(u_\epsilon)}{t} + \epsilon\cdot\parallel v\parallel_a\geq 0$$
Letting $t\rightarrow 0$ it follows that
$<d\Psi_{\lambda}(u_\epsilon),v>+\epsilon\cdot\parallel v \parallel_a
\geq0 $
and we infer that $\parallel d\Psi_{\lambda}(u_\epsilon)\parallel_a \leq\epsilon $.\\
We deduce that there exists a sequence $\{w_n\}\subset B_{\rho}(0)$
such that
\begin{equation}\label{L1770}
  \Psi_{\lambda}(w_n) \longrightarrow \underline{c}\quad \mbox{and} \quad  d\Psi_{\lambda}(w_n)\longrightarrow 0_{X^*}.
\end{equation}

The sequence $\{w_n\}$ is bounded in $X$. Thus, there exists $w$
in $X$ such that, up to a subsequence, $\{w_n\}$ converges weakly to
$w$ in $X$.
Since $\max(r_1(x), r_2(x))<p^{\ast}(x)$, for all $x\in\overline{\Omega}$,
then $X$ is compactly embedded in $L^{r_1(x)}(\Omega)$ and
$L^{r_2(x)}(\Omega)$.
In order to establish the strong convergence of $\{u_n\}$ on $X$, we use proposition \ref{lim} and the following auxiliary property.
\begin{prop}\label{lim1}
We have\\
%$(i)$ $\displaystyle\lim_{n\to
%\infty} \int_{\Omega}V_1(x)|u_{n}|^{q(x)-2}u_{n}(u_{n}-u)dx=0.$\\
 $\displaystyle\lim_{n\to
\infty} \int_{\Omega}V_2(x)|u_{n}|^{\alpha(x)-2}u_{n}(u_{n}-u
)dx=0.$

\end{prop}

\begin{proof}
Using H\"older's inequality (\ref{7}) we have
$$\displaystyle
\int_{\Omega}V_2(x)|u_{n}|^{\alpha(x)-2}u_{n}(u_{n}-u)dx\leq
|V_2|_{s_{2}(x)}||u_n|^{\alpha(x)-2}u_n|_{\frac{\alpha(x)}{\alpha(x)-1}}|u_n-u|_{r_2(x)},$$

Next, by Proposition \ref{norm1},
 if $$\displaystyle ||u_n|^{\alpha(x)-2}u_n|_{\frac{\alpha(x)}{\alpha(x)-1}}>1$$
 then $$\displaystyle
||u_n|^{\alpha(x)-2}u_n|_{\frac{\alpha(x)}{\alpha(x)-1}}\leq |u_n|^{\alpha^+}_{\alpha(x)}.$$
Using the compact embedding $X\hookrightarrow L^{\alpha(x)}(\Omega)$, we conclude the proof of the proposition.
\end{proof}
Since $d\Psi_{\lambda}(w_n)\to  0$ and $\{w_n\}$ is bounded in $X$ we
have
\begin{eqnarray*}
|\langle d\Psi_{\lambda}(w_n),w_n-w\rangle|&\leq&
|\langle d\Psi_{\lambda}(w_n),w_n\rangle|+|\langle d\Psi_{\lambda}(w_n),w\rangle|\\
&\leq&\|d\Psi_{\lambda}(w_n)\|_a\| w_n\|_a+ \| d\Psi_{\lambda}(w_n)\|_a \|
w\|_a.
\end{eqnarray*}
Moreover, by Propositions \ref{lim} and \ref{lim1}, we have
$$
\lim_{n\to \infty}\langle d\Psi_{\lambda}(w_n),w_n-w\rangle=0.
$$
Therefore
\begin{eqnarray*}
\displaystyle\lim_{n\to \infty}M\Big(\int_{\Omega}\frac{1}{p(x)}(|\Delta w_n|^{p(x)}+a(x)|w_n|^{p(x)})dx\Big)\ \int_{\Omega}\Big(|\Delta w_n|^{p(x)-2}\Delta w_n (\Delta w_n-\Delta w)+\\a(x)|w_n|^{p(x)-2}w_n(w_n-w)\Big)dx=0.
\end{eqnarray*}
Combining hypothesis $(H_0)$ and Proposition \ref{111},  we deduce that
$\{w_n\}$ converges strongly to $w$ in $X$.
Since $\Psi_{\lambda}\in C^1(X,\mathbb{R})$, we conclude that
\begin{equation}\label{19}
 d\Psi_{\lambda}(w_n)\rightarrow d\Psi_{\lambda}(w),\ \mbox{as}\ n\rightarrow \infty.
\end{equation}

Now, relations \eqref{L1770} and \eqref{19} yield
\begin{equation}\label{critical}
\Psi_{\lambda}(w)=\underline{c}<0\ \mbox{ and}\ \ d\Psi_{\lambda}(w)=0.
\end{equation}

In order to show that $w$ is a
solution of problem $(\ref{1})$, it remains to show that $\Delta w=0$ on $\partial\Omega$.
Due to relation \eqref{critical}, $w\in X\setminus\{0\}$ is a critical point of $\Psi_\lambda$, so
\begin{equation}\label{star}M(t)\int_\Omega |\Delta w|^{p(x)-2}\Delta w\Delta vdx=\int_\Omega m(x)vdx\quad\mbox{for all}\ v\in X,
\end{equation}
where
$$m(x)=\lambda (V_1(x)|w|^{q(x)-2}w-V_2(x)|w|^{\alpha(x)-2}w)-M(t)a(x)|w|^{p(x)-2}w.$$
Relation (\ref{star}) implies that
\begin{equation}\label{1star}M(t)\int_\Omega |\Delta w|^{p(x)-2}\Delta w\Delta vdx=\int_\Omega m(x)vdx\quad\mbox{for all}\ v\in C^\infty_0(\Omega).\end{equation}

Let $\zeta$ be the unique solution of the problem
$$\left\{\begin{array}{lll}
&\Delta\zeta =m(x)\quad&\mbox{in}\ \Omega\\
&\zeta =0\quad&\mbox{on}\ \partial\Omega.\end{array}\right.
$$
Relation (\ref{1star}) yields
$$M(t)\int_\Omega |\Delta w|^{p(x)-2}\Delta w\Delta vdx=\int_\Omega (\Delta\zeta)vdx\quad\mbox{for all}\ v\in C^\infty_0(\Omega).$$Using the Green formula we have
$$\int_\Omega (\Delta\zeta)vdx=\int_\Omega \zeta\Delta vdx.$$
Therefore
\begin{equation}\label{2star}
M(t)\int_\Omega |\Delta w|^{p(x)-2}\Delta w\Delta vdx=\int_\Omega\zeta \Delta vdx\quad\mbox{for all}\ v\in C^\infty_0(\Omega).\end{equation}

On the other hand, for all $\tilde{w}\in C^\infty_0(\Omega)$ there exists a unique $v\in C^\infty_0(\Omega)$ such that $\Delta v=\tilde{w}$ in $\Omega$. Thus, relation (\ref{2star}) can be rewritten as
$$\int_\Omega (M(t)|\Delta w|^{p(x)-2}\Delta w-\zeta)\tilde{w}dx=0\quad\mbox{for all}\ \tilde{w}\in C^\infty_0(\Omega).$$
Applying the fundamental lemma of the calculus of variations, we deduce that
$$M(t)|\Delta w|^{p(x)-2}\Delta w-\zeta=0\quad\mbox{in}\ \Omega.$$
According to assertion $(H_0)$ and since $\zeta=0$ on $\partial\Omega$, we conclude that $\Delta w=0$ on $\partial\Omega$.
Thus, $w$ is a  nontrivial weak solution of problem
\eqref{1} such that $\Delta w=0$. Since $\Psi_{\lambda}(|w|)=\Psi_{\lambda}(w)$
then problem \eqref{1} has a non-negative solution. The proof is now complete.\quad$\square$



\subsection{Example}
Let $M(t)=\beta+\gamma t$, where $\beta,\gamma$ are positive constants and $t:=\int_{\Omega}\frac{1}{p(x)}(|\Delta u|^{p(x)}dx+a(x)|u|^{p(x)})dx$. We first observe that $$M(t)\geq\beta >0.$$
Taking $\theta=\frac{1}{2}$, we have $$\widehat{M}(t)=\int_{0}^{t}M(s) ds=\beta t+\frac{1}{2}\gamma t^2\geq\frac{1}{2}(\beta+\gamma t)t=(1-\theta)M(t)t.$$
Consider the nonlocal problem
\begin{equation}\label{ex1}
\left\{\begin{array}{lll}M(t)(\Delta^{2}_{p(x)} u+a(x)|u|^{p(x)-2}u)&=\lambda
V_1(x)|u|^{q(x)-2}u,
\quad & x\in\Omega\\
%&u\not\equiv 0,\quad  &\mbox\in\Omega\\
&u=\Delta u=0,\quad   & x\in\partial\Omega,\\
\end{array}\right.
\end{equation}
where $a\in L^{\infty}(\Omega)$ such that ${\rm essinf}_{x\in \Omega}a(x)> 0$, the functions $p, q$ and $V_1$ satisfy hypothesis assertion $(H_2)$. Then, by Ekeland's variational principle, there exists $\lambda^*$ such that any $\lambda\in(0,\lambda^*)$ is eigenvalue for problem (\ref{ex1}). Moreover if we consider the problem
\begin{equation}\label{ex}
\left\{\begin{array}{lll}M(t)(\Delta^{2}_{p(x)} u+a(x)|u|^{p(x)-2}u)&=\lambda(
V_1(x)|u|^{q(x)-2}u-V_2(x)|u|^{\alpha(x)-2}u),
\quad & x\in\Omega\\
%&u\not\equiv 0,\quad  &\mbox\in\Omega\\
&u=\Delta u=0,\quad   & x\in\partial\Omega,\\
\end{array}\right.
\end{equation}
where the functions $p, q$, $\alpha$ and $V_2$ satisfy hypothesis assertion $(H_3)$, then there exists $\lambda_{1}^{*}$ such that any $\lambda\in(0,\lambda_{1}^{*})$ is eigenvalue for problem (\ref{ex}).

\subsection{Final comments}
(i) Problems $\eqref{p1}$ and $\eqref{1}$ correspond to a {\it subcritical} setting, as described by Remark \ref{r1} and Remark \ref{r}. We consider that valuable research directions correspond either to the {\it critical} or to the {\it supercritical} framework (in the sense of Sobolev variable exponents). No results are known even for the {\it almost critical} case with {\it lack of compactness}. More precisely, with the same notations as in  Remark \ref{r1} and \ref{r}, a very interesting open problem is to study the qualitative analysis of solutions of problem $\eqref{r1}$, provided that
there exists $z_1\in\Omega$ such that
 $$\max({r_1(z_1),s_1'(z_1)q(z_1)})=p^*(z_1),$$ but
$$\max({r_1(x),s_1'(x)q(x)})<p^{*}(x)\quad\mbox{for all}\ z\in\Omega\setminus\{z_1\},$$
and those of problem $\eqref{1}$ provided that
 there exists $z_1,\, z_2\in\Omega$ such that
 $$\max({r_1(z_1),s_1'(z_1)q(z_1)})=p^*(z_1)\quad \mbox{and}\quad \max(r_2(z_2)s_2'(z_2)\alpha(z_2))=p^*(z_2)$$ but
 $$\max({r_1(x),s_1'(x)q(x)})<p^{*}(x)\quad\mbox{for all}\ z\in\Omega\setminus\{z_1\}$$
 and $$\max(r_2(x)s_2'(x)\alpha(x))<p^{*}(x)\quad\mbox{for all}\ z\in\Omega\setminus\{z_2\}.$$

(ii)  Another very interesting research direction is to extend the approach developed in this paper to the abstract setting recently studied by Mingione {\it et al.} \cite{mingi1, mingi02, mingi2}, namely{\it double phase} problems with associated energies of the type
$$u\mapsto \int_\Omega [|\Delta u|^{p_1(x)}+V(x)|\Delta u|^{p_2(x)}]dx$$
and
$$u\mapsto \int_\Omega [|\Delta u|^{p_1(x)}+V(x)|\Delta u|^{p_2(x)}\log(e+|x|)]dx,$$
where $p_1(x)\leq p_2(x)$, $p_1\not=p_2$, and $V(x)\geq 0$. Considering two different materials with power hardening exponents $p_1(x)$ and $p_2(x)$ respectively, the coefficient $V(x)$ dictates the geometry of a composite of the two materials. When $V(x)>0$ then $p_2(x)$-material is present, otherwise the $p_1(x)$-material is the only one making the composite.

These problems extend to a bi-harmonic setting with variable exponents the pioneering papers by Marcellini \cite{marce1, marce2} on $(p,q)$-problems, which involve integral functionals of the type
$$u\mapsto\int_\Omega F(x,\nabla u)dx.$$
The integrand $F:\Omega\times{\mathbb R}^N\rightarrow {\mathbb R}$ satisfied unbalanced polynomial growth conditions of the type
$$|\xi|^p \lesssim F(x,\xi)\lesssim |\xi|^q\quad\mbox{with $1<p<q$,}$$
for every $x\in\Omega$ and $\xi\in{\mathbb R}^N$.

(iii) We suggest to extend the methods developed in this paper to the more general framework of Musielak-Orlicz-Sobolev spaces (see \cite[Chaper 4]{radrep} for a collection of stationary problems studied in these function spaces).

\subsection*{Acknowledgements} The authors thank an anonymous referee  for a careful and constructive analysis of this paper and for the remarks and comments, which have considerably improved the
initial version of the present work.
V.D.~R\u adulescu acknowledges the support through a grant  of the Romanian Ministry
of Research and Innovation, CNCS--UEFISCDI, project number PN-III-P4-ID-PCE-2016-0130,
within PNCDI III.









\end{document}
